Author’s response to reviews

Title: Prescribing errors in electronic prescriptions for outpatients intercepted by pharmacists and the impact of prescribing workload on error rate in a Chinese Tertiary-care Women and Children's Hospital

Authors:

Jianhui Yang (814328154@qq.com)
Yu-fang Liao (54698130@qq.com)
Wu-bin Lin (18094886@qq.com)
Wen Wu (990666272@qq.com)

Version: 3 Date: 14 Dec 2019

Author’s response to reviews:

1. Image Attribution
   -- Please confirm whether the images depicted in [FIGURE 1] are your own or taken from another source.
   If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it.
   If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time.
   Should you require an alternative source you may wish to try Wikimedia Commons:
   http://commons.wikimedia.org/wiki/Main_Page

   Response: I confirm that Figure 1 was created all by ourselves.

2. Remove Files
   -- Please remove the [renamed] response to reviewers’ comments and [STROBE] checklist supplementary files from your file inventory, as these are no longer required at this stage of the editorial process.

   Response: We had removed the files you pointed.

3. Clean Manuscript
   -- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional
files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Response: The manuscript of last submission was a clean version.